Author's response to reviews

Title: Screening of crude extracts of six medicinal plants used in South-West Nigerian unorthodox medicine for anti-methicillin resistant Staphylococcus aureus activity.

Authors:

Kabir’ O Akinyemi (akinyemiko@yahoo.com)
Olukayode Oladapo (oladapokayode2005@yahoo.com)
Okwara E Clement (okwaraclement@yahoo.com)
Christopher E Ibe (ibecharis2002ng@yahoo.com)
Kehinde .A Fasure (fasureak@yahoo.com)

Version: 4 Date: 4 February 2005

Author’s response to reviews: see over
Response to referee’s comment
1. The manuscript has been proof–read and edited by our secretariat.
2. All the grammatical and spelling mistakes have been adequately addressed.
3. The terminology MRSA has now been chosen for consistency.
4. Inconsistencies pointed out in the table format, front sizes, bolding of words etc have been addressed.
5. Answer to question on immunomodulation: Two medicinal plants that were inactive in vitro against MRSA in this study referred. Since it has been documented in literature that the fact that a particular plant is inactive against a pathogen in vitro may not totally preclude the plant from performing other functions in the host system, one of such functions is immunomodulation. Although, we did not perform immunomodulating action on these inactive plant extracts as clearly stated in our revised discussion. We only suggested that, to corroborate what obtains in literature.
6. All other minor corrections had also been effected.